%**	This is a pure LaTeX 2e file [article class]
%**	Filename:	.tex
%**	Last update:
%**     Final version:
%**	Title:
%**	Purpose:

%**	Author (and responsible for all errors):

%***************************************************************************%
%**                                                                       **%
%**	 Article style of BOZHIDAR ZAKHARIEV ILIEV: 		 	  **%
%**                                                                       **%
%**	 Laboratory of Mathematical Modeling in Physics			  **%
%**	 Institute for Nuclear Research and Nuclear Energy                **%
%**	 Bulgarian Academy of Sciences                                    **%
%**	 Boul. Tzarigradsko chauss\'ee~72, 1784 Sofia, Bulgaria           **%
%**	 E-mail address: bozho@inrne.bas.bg				  **%
%**      URL: http://theo.inrne.bas.bg/~bozho/	        		  **%
%**                                                                       **%
%**                                                                       **%
%**			(May be used freely by everybody)                 **%
%**                                                                       **%
%***************************************************************************%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                                                       %%
%% 		Beginning of \CharacterTable                             %%
%%                                                                       %%
%%  {Upper-case    \A\B\C\D\E\F\G\H\I\J\K\L\M\N\O\P\Q\R\S\T\U\V\W\X\Y\Z  %%
%%   Lower-case    \a\b\c\d\e\f\g\h\i\j\k\l\m\n\o\p\q\r\s\t\u\v\w\x\y\z  %%
%%   Digits        \0\1\2\3\4\5\6\7\8\9                                  %%
%%   Exclamation   \!     Double quote  \"     Hash (number) \#          %%
%%   Dollar        \$     Percent       \%     Ampersand     \&          %%
%%   Acute accent  \'     Left paren    \(     Right paren   \)          %%
%%   Asterisk      \*     Plus          \+     Comma         \,          %%
%%   Minus         \-     Point         \.     Solidus       \/          %%
%%   Colon         \:     Semicolon     \;     Less than     \<          %%
%%   Equals        \=     Greater than  \>     Question mark \?          %%
%%   Commercial at \@     Left bracket  \[     Backslash     \\          %%
%%   Right bracket \]     Circumflex    \^     Underscore    \_          %%
%%   Grave accent  \`     Left brace    \{     Vertical bar  \|          %%
%%   Right brace   \}     Tilde         \~}                              %%
%%                                                                       %%
%% 		End of \CharacterTable                                   %%
%%                                                                       %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\typeout{}
\typeout{????????????????????????????????????????????????????????????????}
\typeout{}
\typeout{This is the file of the article}
\typeout{" "}
\typeout{by Bozhidar Zakhariev Iliev}
\typeout{Its initial draft version was written during the period}
\typeout{???-- ???}
\typeout{}
\typeout{vvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvvv}
\typeout{}

%====>>	BEGINNING OF EXTERNAL FILES	====>>	====>>	====>>	====>>

% Beginning of the file
%\begin{filecontents}{}
%\end{filecontents}
% End of the file

% Beginning of the bibliography file .bbl
%\begin{filecontents}{.bbl}
%\end{filecontents}
% End of the bibliography file .bbl


% Beginning of the file Bozho-LaTeX-Style.sty
\begin{filecontents}{Bozho-LaTeX-Style.sty}
\end{filecontents}
% End of the file Bozho-LaTeX-Style.sty


% Beginning of the file Bozho-LaTeX-Logo.sty
\begin{filecontents}{Bozho-LaTeX-Logo.sty}
\end{filecontents}
% End of the file Bozho-LaTeX-Logo.sty

% <<====	END OF EXTERNAL FILES	<<====	<<====	<<====	<<====



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% -->-->-->-->>
%	Beginning of the PREAMBLE  of the BOZHO'S article style    -->-->-->>
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% -->-->-->-->>


\documentclass[11pt,titlepage,a4paper]{article}
%
\usepackage{bozhomac}	% loads: full AMSLaTeX and AMSFonts
\usepackage{bozhlogo}	% the logo of BOZHO commands \BOZHO and \bozho
% !!!!!!!!!!!!	 when sending files - input the files themselves here!!!
%	other \usepackage commands
\usepackage{showkeys}	% for draft viewing and printing
\usepackage{cite}	% for condensing sequential cited works
% \usepackage{hyperref}	% for inclusion of hyperlink info in TeX and LaTeX docs

% \includeonly{empty,	% for temporary typesetting of the current text
%tmp			% filename, NO extension, of the temporary file
%}			% it must be the same as in the \include command



%		formatting the TITLE page	>========>
%
\title{\bfseries	\vspace*{-1.678902345in}
{\huge           }% main title
% \\ \vspace{0.22ex} \\	\vspace{1.1ex}	{\LARGE     }% subtitle
}

\vspace{1.7ex}

\author{
Bozhidar Z.\ Iliev
\thanks{Laboratory of Mathematical Modeling in Physics,
Institute for Nuclear Research and \mbox{Nuclear} Energy,
Bulgarian Academy of Sciences,
Boul.\ Tzarigradsko chauss\'ee~72, 1784 Sofia, Bulgaria}
\thanks{E-mail address: bozho@inrne.bas.bg}
\thanks{URL: http://theo.inrne.bas.bg/$\sim$bozho/}
}


% 	Put bellow any additional title page info
%	and (partially) remove it when sending the file
%
%
\date{
 \vspace{2.27ex}\ShortTitle{}	\\[0.27ex]
 \vspace{3.27ex}
\small
%
	\begin{tabular}{r@{$\colon\to~$}l}
% \vspace{0.09ex} Basic ideas	&	\\[0.09ex]
 \vspace{0.09ex} Began		&	\\[0.09ex]
 \vspace{0.09ex} Ended		&	\\[0.09ex]
%
%\vspace{0.09ex} Initial typeset& \today	\\[0.09ex]
%
% \vspace{0.09ex} Revised	&		\\[0.09ex]
% \vspace{0.09ex} Last update	& \today	\\[0.09ex]
 \vspace{0.27ex} Produced	& \fbox{\today}	\\[0.27ex]
	\end{tabular} \\[1.27ex]
%
 \small
	\begin{tabular}{r@{$\colon~$}l}
\normalsize\sffamily\bfseries
 \vspace{0.27ex} http://arXiv.org e-Print archive No. &
\normalsize\sffamily\bfseries
?????/??? \\[1.27ex]
% \vspace{0.27ex} Submitted to		&	\\[0.27ex]
% \vspace{0.27ex} Resubmitted to	&	\\[0.27ex]
% \vspace{0.27ex} Accepted in		&	\\[0.27ex]
% \vspace{0.27ex} Published in		&	\\[0.27ex]
	\end{tabular} \\[-0.27ex]
\normalsize
%
 \vspace{4.27ex}{\Huge\BOZHO}	\\[4.27ex]
%
% \vspace{0.27ex}\Subject{?}	\\[2.27ex]
	\begin{tabular}{r@{\hspace{0.512em}}|@{\hspace{0.512em}}l}
% \vspace{0.27ex}\MSC[2010]{?}	%	\\[0.27ex]
&
% \vspace{0.27ex}\PACS[2010]{?} %	\\[0.27ex]
	\end{tabular} \\[1.27ex]
% \vspace{0.27ex}\KeyWords{?}	\\[0.27ex]
}


%		End of the title page           <========<




\listfiles			% lists all of the files being used


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%      Specific settings for this work only      %%%%%%%%%%%%%%
%%%%%%%%								%%%%%%
%+++++++>>>	Beginning of page layout	+++++++>>>
\pagestyle{myheadings}
\markright{\underline{\itshape\bfseries Bozhidar Z. Iliev:
	\upshape\sffamily\bfseries  PAPER's TITLE}}

%	For maximal and reasonable usage of the a4 paper:
 \topmargin -16.5mm 			%-16mm
 \addtolength{\textheight}{34mm}	% {46mm}
 \addtolength{\textwidth}{1.10in}	% {46mm}
 \oddsidemargin -0.05in			% -9mm
 \evensidemargin -0.05in		% -9mm
%-------<<<	End of page layout		-------<<<

%%%%%%%%								%%%%%%
%%%%%%%%%%%%%%%%    End of the specific settings for this work   %%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% <<--<--<--<--<--<--<
%	End of the PREAMBLE  of the BOZHO'S article style   <<--<--<--<--<--<
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% <<--<--<--<--<--<--<



\begin{document}		% BEGINNING OF THE DOCUMENT

\renewcommand{\thepage}{\roman{page}}

\renewcommand{\thefootnote}{\fnsymbol{footnote}} % special footnote symbols
\maketitle				% the title (page) is put here
\renewcommand{\thefootnote}{\arabic{footnote}}   % usual footnote symbols


\tableofcontents		% the table of contents is put here



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%									%%%%%
%%%%%		actual beginning of the document			%%%%%
%%%%%									%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%






%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->---->-->-->>
\begin{abstract}


\end{abstract}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<

\renewcommand{\thepage}{\arabic{page}}

% \include{tmp} \file with temporary/current typesetting text



%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->---->-->-->>
\section {Introduction}
\label{Introduction}
%			BEGINNING OF SECTION~\ref{Introduction}






%			END OF SECTION~\ref{Introduction}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<




%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section {}
\label{Sect2}
%			BEGINNING OF SECTION~\ref{Sect2}






%			END OF SECTION~\ref{Sect2}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<



%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section {}
\label{Sect3}
%			BEGINNING OF SECTION~\ref{Sect3}






%			END OF SECTION~\ref{Sect3}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<



%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section {}
\label{Sect4}
%			BEGINNING OF SECTION~\ref{Sect4}






%			END OF SECTION~\ref{Sect4}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<



%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section {}
\label{Sect5}
%			BEGINNING OF SECTION~\ref{Sect5}






%			END OF SECTION~\ref{Sect5}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<




%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section {}
\label{Sect6}
%			BEGINNING OF SECTION~\ref{Sect6}






%			END OF SECTION~\ref{Sect6}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<



%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section {Conclusion}
\label{Conclusion}
%			BEGINNING OF SECTION~\ref{Conclusion}






%			END OF SECTION~\ref{Conclusion}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<



%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section*{Acknowledgments}
%			BEGINNING OF SECTION~\ref{Acknowledgments}







%			END OF SECTION~\ref{Acknowledgments}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<







% >>>>====>>>>	 BEGINNING OF APPENDICES	====>>>>	====>>>>
\appendix

% Use the following for only 1 Appendix in the paper
%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section*{Appendix}

\renewcommand{\thesection}{A}
\addcontentsline{toc}{section}{Appendix}
\label{Appendix}
%			BEGINNING OF SECTION~\ref{Appendix}




%			END OF SECTION~\ref{Appendix}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<




% Use the following for more then one Appendices in the paper
% The table of contents title must be given without space in [] just before ]
\renewcommand{\thesection}{Appendix~\Alph{section}}

%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\section[\hspace*{10.1ex}]{}

	\renewcommand{\thesection}{\Alph{section}}
	\label{AppendixA}
%			BEGINNING OF SECTION~\ref{AppendixA}





%			END OF SECTION~\ref{appendixA}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<



%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
\renewcommand{\thesection}{Appendix~\Alph{section}}

\section[\hspace*{11.1ex}]{}

	\renewcommand{\thesection}{\Alph{section}}
	\label{AppendixB}
%			BEGINNING OF SECTION~\ref{AppendixB}





%			END OF SECTION~\ref{appendixB}
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<

% <<<<==== END OF APPENDICES	<<<<====	<<<<====<<<<





%-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->-->>
%			BEGINNING OF BIBLIOGRAPHY
\addcontentsline{toc}{section}{References}
\bibliography{bozhopub,bozhoref}
\bibliographystyle{unsrt}
\addcontentsline{toc}{subsubsection}{This article ends at page}
%\addtocontents{toc}{}
%			END OF BIBLIOGRAPHY
%<<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<--<


%           END OF subsection
%<<<<<-----------------------------------------------------------------------



%>>>>>-----------------------------------------------------------------------
%           BEGINNING OF subsection



\end{document}

>=>=>=>=>=>
 	   PUT HERE ANY ADDITIONAL APPLICABLE INFORMATION

							 <=<=<=<=<=<=<=<=<=<

=============================================================================
=============================================================================
>>>		I do not want to profit from the Science,                 <<<
>>		but the Science may profit from me!                        <<
>						    Bozho                   <
=============================================================================
=============================================================================

% arXiv HAT
\\
Title:
Authors: Bozhidar Z. Iliev (Institute for Nuclear Research and Nuclear
	Energy, Bulgarian Academy of Sciences, Sofia, Bulgaria)
Comments: ?? LaTeX pages. The packages AMS-LaTeX and amsfonts are required.
	For other papers on the same topic, view
	http://theo.inrne.bas.bg/~bozho/
Report-No:
MSC-class: ...  (Primary) ...  (Secondary)
\\
$ Abstract (Delete this raw!!)
\\
